Author's response to reviews

Title: Endothelial Dysfunction in Obese Non-Hypertensive Children Without Evidence of Sleep Disordered Breathing

Authors:

Rakesh Bhattacharjee (rakesh.bhattacharjee@sickkids.ca)
Wadha H Alotaibi (waltobaib@yahoo.com)
Leila Kheirandish-Gozal (igozal@peds.bsd.uchicago.edu)
Oscar Sans Capdevila (osans@hsjdbcn.org)
David Gozal (dgozal@peds.bsd.uchicago.edu)

Version: 4 Date: 8 February 2010

Author's response to reviews:

Dear Dr Clark

A sentence indicating the approval of this study by the IRB and the appropriate process of obtaining informed consent has now been included in the manuscript.

Best Regards

David Gozal, MD